Author’s response to reviews

Title: Asymmetry and Activity Indices and Occlusal Contacts Distribution in Orthodontically Treated versus Nontreated Young Adults.

Authors:

Aneta Wieczorek (aneta.wieczorek@uj.edu.pl)
Jolanta Loster (jolanta.loster@uj.edu.pl)

Version: 2 Date: 29 October 2014

Author’s response to reviews:

Kraków 29 October, 2014

BMC Oral Health

Dear Sirs,

Ad. 1 Title page
It contain, the names, institutions, country and email addresses of all authors, and the full postal address of the submitting author

Ad. 2
Line Numbering.
We includes line and page numbers

Ad. 3
Please include list of abbreviations.
List of abbreviations is included.

Ad. 4
Availability of Supporting Data section
Don’t concern

Sincerely,

Aneta Wieczorek, DDS, PhD